Author’s response to reviews

Title: The diagnostic performance of $^{18}$F-FAMT PET and $^{18}$F-FDG PET for malignancy detection: A meta-analysis

Authors:

Arifudin Achmad (arifudinachmad@gmail.com)
Anu Bhattarai (bhattaraianu28@gmail.com)
Ryan Yudistiro (ryannuclear@gmail.com)
Yusri Heryanto (yusri_dh@gmail.com)
Tetsuya Higuchi (tetsuyah92md@gmail.com)
Yoshito Tsushima (yoshitotsushima@gunma-u.ac.jp)

Version: 2 Date: 11 Dec 2017

Author’s response to reviews:

COVER LETTER – MINOR REVISION

General Response to Editors and Reviewers:

Dear the respected BMC Medical Imaging Editors and reviewers,

Thank you for reviewing this manuscript. The authors had addressed each of the comments as described below. The individual contributions of all authors to the manuscript are now clearly specified in the Authors’ Contributions section.

Best Regards,
Arifudin Achmad

Response to Reviewers (Q: Question, R: Response)

Reviewer 2 (Dr. Ali Azhdarinia):

The authors have effectively addressed the prior concerns. Two minor suggestions are listed below:

Q1. Page 3, line 30: "and to rule out benign lesions in lesions suspected to be malignant neoplasms" should be revised?
R1. Revised to: “and to rule out benign lesions from malignant neoplasms”.

Q2. Page 13, line 28: replace "excellence" with "effectiveness"

R2. Revised as recommended.